Author’s response to reviews

Title: Intention, magnitude and factors associated with bottle feeding among mothers of 0-23 months old children in Holeta town, Central Ethiopia: A cross sectional study

Authors:

Tadesse Bedane (tadeek2011@yahoo.com)

Hirut Cherie (hirutas2000@yahoo.com)

Version: 1 Date: 17 Jan 2017

Author’s response to reviews:

1. Please note, if your manuscript is accepted you will not be able to make any changes to the authors, or order of authors, of your manuscript once the editor has accepted your manuscript for publication. If you wish to make any changes to authorship before you resubmit your revisions, please reply to this email and ask for a 'Request for change in authorship' form which should be completed by all authors (including those to be removed) and returned to this email address. Please ensure that any changes in authorship fulfil the criteria for authorship as outlined in BioMed Central's editorial policies

-----------------------------------------------------

RESPONSE : all authors were included in the manuscript as BioMed Central's editorial policies authorship criteria

-----------------------------------------------------

2. Editor's comments:

Dear Authors,

Thank you for submitting your paper to BMC Nutrition.

It addresses an important and interesting topic; however, before it can initiate the peer review process the methods section has to be implemented. It has to be described in more details the questionnaire that has been used, when the interviews have been made, who did them ect. It would also be interesting to differentiate mothers giving their own milk form those giving formula since the reasons may be different.

-----------------------------------------------------
RESPONSE :methodology part is edited accordingly. We identified what type of foods mothers were giving by bottle in the questionnaire, but we didn't get mothers giving their own milk. The type of food used is found in Table 3 (result part).

---------------------

3. Editorial Policies

---------------------

Please read the following information and revise your manuscript as necessary. If your manuscript does not adhere to our editorial requirements, this will cause a delay whilst the issue is addressed. Failure to adhere to our policies may result in rejection of your manuscript.

In accordance with BioMed Central editorial policies and formatting guidelines, all submissions to BMC Nutrition must have a Declarations section which includes the mandatory sub-sections listed below. Please refer to the journal's Submission Guidelines web page for information regarding the criteria for each sub-section (http://www.biomedcentral.com/bmcnutr).

Where a mandatory section is not relevant to your study design or article type, for example, if your manuscript does not contain any individual persons data, please write "Not applicable" in these sections.

For the 'Availability of data and materials' section, please provide information about where the data supporting your findings can be found. We encourage authors to deposit their datasets in publicly available repositories (where available and appropriate), or to be presented within the manuscript and/or additional supporting files. Please note that identifying/confidential patient data should not be shared. Authors who do not wish to share their data must state that data will not be shared, and provide reasons for this in the manuscript text. For further guidance on how to format this section, please refer to BioMed Central's editorial policies page - http://www.biomedcentral.com/submissions/editorial-policies#availability+of+data+and+materials.

Declarations

- Ethics approval and consent to participate
- Consent to publish
- Availability of data and materials
- Competing interests
- Funding
- Authors' Contributions
- Acknowledgements
- Authors' Information

RESPONSE: Done accordingly (we put declaration at the last page of manuscript next to conclusion and recommendation)